Author's response to reviews

**Title:** Incidence of Human Herpesvirus 8 (HHV-8) Infection among HIV-uninfected individuals at high risk for sexually transmitted infections

**Authors:**

Massimo Giuliani (giuliani@iss.it)
Paola Cordiali Fei (cordiali-fei@ifo.it)
Concetta Castilletti (castilletti@inmi.it)
Aldo Di Carlo (dicarlo@ifo.it)
Guido Palamara (mst.hiv@ifo.it)
Stefano Boros (boros@iss.it)
Giovanni Rezza (g.rezza@iss.it)

**Version:** 6  **Date:** 3 December 2007

**Author's response to reviews:** see over
To
The BioMed Central Editorial Production Team
BioMed Central
Middlesex House
32-42 Cleveland Street
London
W1T 4LB
United Kingdom

Rome, December 3, 2007

Reference: MS: 3452481921244804 – “Incidence of Human Herpesvirus 8 (HHV-8) Infection Among HIV-Uninfected Individuals at High Risk for Sexually Transmitted Infections”.

Dear pre-core accept Team:

I have attached the final version of manuscript no. 3452481921244804, entitled, “Incidence of Human Herpesvirus 8 (HHV-8) Infection Among HIV-Uninfected Individuals at High Risk for Sexually Transmitted Infections”, modified in according to your formatting requested changes.

In particular in the manuscript we have changed the points specified below:

a) In the title all unnecessary capitals have been removed.
b) The affiliations have been modified and numbered.
c) The running title and the word count have been deleted.
d) In the “section headings” all unnecessary capitals have been removed.
e) In the Competing interests the initials AM have been deleted.
f) In the end of the Authors’ contributions, the phrase “All authors read and approved...” has been added.
g) All references have been completed with all authors.
h) The table #4 is now mentioned in the Results section.
i) All references have been cited in the text before the full stop and in square brackets.

Sincerely,

Dr. Massimo Giuliani, DSc. (Corresponding author)
Reparto AIDS e Malattie Sessualmente Trasmesse
Istituto Superiore di Sanità
00161 Rome, Italy
phone: +39 06 49387211-13
fax: +39 06 49387210
e-mail: giuliani@iss.it